<u>Evaluating process and effectiveness of a low-intensity CBT intervention for women with</u> gynaecological cancer (the EPELIT Trial)

Nicholas J. Hulbert-Williams*, Lee Hulbert-Williams, Ryan James Flynn, Rosina Pendrous, Carey MacDonald-Smith, Anna Mullard, Brooke Swash, Gemma Evans & Annabel Price

*corresponding author

Author contact details:

N J Hulbert-Williams (<u>n.hulbertwilliams@chester.ac.uk</u>): Centre for Contextual Behavioural Science, School of Psychology, University of Chester, Parkgate Road, Chester, CH1 4BJ, UK.

L Hulbert-Williams (<u>I.hulbertwilliams@chester.ac.uk</u>): Centre for Contextual Behavioural Science, School of Psychology, University of Chester, Parkgate Road, Chester, CH1 4BJ, UK.

R J Flynn (ryan.j.flynn93@gmail.com): Centre for Contextual Behavioural Science, School of Psychology, University of Chester, Parkgate Road, Chester, CH1 4BJ, UK.

R Pendrous (<u>ropendrous@gmail.com</u>): Centre for Contextual Behavioural Science, School of Psychology, University of Chester, Parkgate Road, Chester, CH1 4BJ, UK.

C MacDonald-Smith (<u>carey.macdonald-smith@wales.nhs.uk</u>): North Wales Cancer Treatment Centre, Glan Clwyd Hospital, Sarn Lane, Bodelwyddan, Rhyl, LL18 5UJ, UK.

A Mullard (<u>Anna.Mullard@wales.nhs.uk</u>): Ysbyty Gwynedd, Penrhosgarnedd, Bangor, Gwynedd, LL57 2PW.

B Swash(<u>b.swash@chester.ac.uk</u>): School of Psychology, University of Chester, Parkgate Road, Chester, CH1 4BJ, UK.

G Evans (<u>gemma.evans@chester.ac.uk</u>): School of Psychology, University of Chester, Parkgate Road, Chester, CH1 4BJ, UK.

A Price (ap806@medschl.cam.ac.uk): Department of Psychological Medicine, S2 Outpatient Department, Addenbrooke's Hospital, Cambridge, CB2 0QQ, UK.

Abstract

Background: Improving survival from gynaecological cancers is creating an increasing clinical challenge for long-term distress management. Psychologist-led interventions for cancer survivors can be beneficial, but are often costly. The rise of the Psychological Wellbeing Practitioner (PWP) workforce in the UK might offer a cheaper, but equally effective, intervention delivery method that is more sustainable and accessible. We aimed to test the effectiveness of a PWP co-facilitated intervention for reducing depression and anxiety, quality of life and unmet needs.

Methods: We planned this trial using a pragmatic, non-randomised controlled design, recruiting a comparator sample from a second clinical site. The intervention was delivered over six-weekly sessions; data were collected from participants at baseline, weekly during the intervention, and at one-week and three-month follow-up. Logistical challenges meant that we only recruited 8 participants to the intervention group, and 26 participants to the control group.

Results: We did not find significant, between-group differences for depression, quality of life or unmet needs, though some differences at follow-up were found for anxiety (p<.001). Analysis of potential intervention mediator processes indicated the potential importance of self-management self-efficacy. Low uptake into the psychological intervention raises questions about (a) patient-driven needs for group-based support, and (b) the sustainability of this intervention programme.

Conclusions: This study failed to recruit to target; the under-powered analysis likely explains the lack of significant effects reported, though some trends in the data are of interest. Retention in the intervention group, and low attrition in the control group indicate acceptability of the intervention content and trial design; however a small baseline population rendered this trial infeasible in its current design. Further work is required to answer our research questions, but also, importantly, to address low uptake for psychological interventions in this group of cancer survivors.

Trial registration: ClinicalTrials.gov, NCT03553784 (registered 14 June 2018).

Keywords: Gynaecological Cancer – Unmet Needs – Distress – Anxiety – Depression – Low-Intensity – Cognitive Behavioural Therapy – Quality of Life

Plain Language Summary

An increasing number of people are surviving for longer time periods following treatment for gynaecological cancer and this means we need to change how we care for and support a growing cancer survivor population. Psychological distress and poor quality of life are common in people affected by cancer, and these do not always improve once treatment ends. Providing psychological support can be expensive, which means that not everyone who wants it can access it. Psychological Wellbeing Practitioners (PWPs) have been introduced in UK health care. This workforce might offer an alternative for providing psychological support to a greater number of cancer survivors.

We aimed to test how good a PWP co-delivered intervention is at improving depression, anxiety and quality of life in people who had been treated for gynaecological cancer. The intervention was delivered to small groups of patients over six weekly sessions. We compared those who received the intervention with a similar patient group who did not have access to the same psychological support from a different hospital. Participants reported their psychological wellbeing and quality of life at the point of recruitment, weekly for six weeks, and then at follow-up time-points one week and three months later.

Because of low interest in the group intervention we did not recruit to target. Only 8 participants took part in the intervention, and we recruited only 26 participants in the control group. This means we can't have full confidence in our results. Nonetheless, the findings indicate that this intervention was helpful for improving participants' anxiety levels.

Further trials, which recruit a larger number of cancer survivors, are needed to answer our research questions. However, this trial indicates acceptability and potential benefit. We also need to undertake research to understand why so few cancer survivors wanted to take part in this group-based intervention.

Introduction

Gynaecological cancers (cervical, ovarian, uterine, vaginal and vulvar) accounted for 5.9% (18,026) of all England-based cancer registrations in 2017 [1]. Many gynaecological cancers are associated with improving survival rates [2]; however, ovarian cancer, has one of the smallest increases in one- and five-year survival rates compared to other cancer types [3]. Delay to diagnosis from first symptom presentation is often longer in cervical cancer than other more common cancers [4] which may be contributory to comparatively smaller increased in overall survival [5]. Greater diagnostic delay can also lead to psychological distress [6], an important outcome which is understood to be predictive of survival in some cancer groups [7]. Information on the unmet psychological and supportive care needs of women with gynaecological cancer is limited, particularly comparisons between sub-types of gynaecological diagnosis, and especially in relation to concurrent psychological distress [8.9]. Recent research suggests increased fear of recurrence [10] and sexuality needs [11] may be particularly problematic for this group. Beesley et al's systematic review [12] suggested that psychoeducation may be helpful in supporting gynaecological cancer survivors to manage some late treatment effects; however, more data is needed on the beneficial impacts of psychological treatment on quality of life in people affected by gynaecological cancer [13].

Increasing evidence suggests that psychosocial interventions are beneficial in reducing distress in adults with cancer [14]. Cognitive Behavioural Therapy (CBT) currently enjoys the strongest evidence-base for psychological distress in people affected by cancer [15] including interventions which aim to improve anxiety, depression and quality of life [16,17]. Concerns remain regarding the generalisability of findings due to small sample sizes used; Xiao et al [17] and Hulbert-Williams et al [15], for example, emphasise the need for further well-designed and better powered randomised controlled trials to improve the evidence base for psychological interventions in those affected by cancer.

In gynaecological cancers specifically, the evidence is somewhat less conclusive. Beesley et al.'s systematic review, concluded that counselling-based approaches improve psychological wellbeing for women with gynaecological cancer [18]. Further, Rost et al.'s comparison of Acceptance and Commitment Therapy (ACT) and CBT for women with advanced ovarian cancer reported improved mood and quality of life in both groups [19]. Other studies have shown that interventions to increase physical activity can reduce fatigue [20] and when combined with CBT, improve quality of life in addition to physical health [21]. Mindfulness-based interventions have shown promising evidence for improved sexual functioning in those with ovarian cancer [22]. Null findings for the evidence of psychological intervention in this group have also been reported, however. For example, Chan et al.'s randomised controlled trial of CBT-based psychoeducation, stress management, relaxation, and

pain and distress management reported no significant effect on quality of life or distress compared with treatment as usual control group receiving routine medical care [23]. Chan et al. postulate that the lack of significant effects may have resulted from the decision not to screen participants for high distress which may have diluted the effect of the intervention, and other authors have noted this as a broader methodological problem in psychosocial oncology intervention science [15].

To maximise potential for implementation into standard care, there is a need for trials of psychological interventions that demonstrate not only evidence for improved outcomes, but also cost- and service-effectiveness too [24]. Traditional psychological interventions are typically delivered by a small number of trained therapists [25] and tend to be time-intensive [26]. The cost of training, the shortfall of qualified therapists, and the increasing demand for psychological interventions that comes with improved cancer survival rates are making the current service delivery model potentially untenable for long-term sustainability. As such, there a need for the development of more cost-effective delivery models, including those which can be delivered by non-psychologist members of the existing healthcare team.

For some years, the English National Health Service (NHS) has provided the delivery of brief psychological services by a specially trained workforce of psychological wellbeing practitioners (PWPs) as part a stepped-care model within the Improving Access to Psychological Therapies (IAPT) initiative [27]. Although some questions remain about the overall effectiveness of PWPs [28], this model is attractive because of cost-effectiveness potential: by providing lower-intensity, briefer interventions before symptoms escalate in severity, PWPs can have higher caseloads allowing increased access treatment [29]. Training is also less exhaustive than for other psychological specialists offering potential cost-savings there too. The 2016 NHS Five Year Forward View for Mental Health [30], suggests that two-thirds of the increase in target access figures for IAPT will be in individuals with co-morbid physical and mental health conditions. This has acted as a catalyst to expand the PWP role to include service provision for people with chronic health conditions, including cancer.

An additional cost-benefit element of IAPT services is achieved because of the option for a groupbased delivery model [29], which is also commonplace in cancer care, including in the delivery of CBT-based psychological interventions [31,32]. A recent systematic review of psychotherapeutic interventions for women with metastatic breast cancer concluded that whilst group-based interventions have the strongest evidence or efficacy, they also suffer from some of the lowest uptake and adherence [33]. Outside of cancer care, research has compared individual versus group delivery of CBT in chronic health settings, finding that group-based delivery leads to greater treatment improvement *and* satisfaction [34]. Qualitative evaluation of a psychoeducation and counselling group intervention for post-treatment gynaecological cancer survivors suggests that positive impacts were related to the "... special community of mutual understanding and belonging" that is created [35].

Study aims

Stepped-care models such as that used in the IAPT service are recognised as essential parts of good clinical psychosocial cancer care [36], but given the changing context in UK-based mental health care, research which explores the efficacy of PWP-supported service delivery on cancer outcomes is needed. Our study reports on a trial of a group-based, CBT intervention for gynaecological cancer survivors that was designed around incorporation of PWPs as co-facilitators. Our primary aim was to explore effectiveness in improving depression and anxiety against a non-randomised control group. As secondary aims, we wanted to test: (a) intervention effects on quality of life and unmet needs, and (b) potential mediators of intervention outcomes.

Although originally planned as a full clinical trial, recruitment difficulties at both our intervention and control site meant that we had to stop the trial short of our recruitment targets. Our paper reports both our originally intended design and adjustments made in relation to our recruitment difficulties. We include discussion of learning points related to this which should be considered in planning similar trails elsewhere.

Methods

This trial has been registered at ClinicalTrials.gov (<u>NCT03553784</u>; registered 14 June 2018). The full trial protocol is available on the Open Science Framework (DOI 10.17605/OSF.IO/YGHNE).

Design and setting

This study employed a non-randomised controlled trial design. The intervention group were recruited from a gynaecology-specific psycho-oncology service in England, where the intervention is delivered as part of standard care. A pragmatic non-randomised treatment-as-usual control group were recruited from a different clinical setting (North Wales). A fully randomized design was not possible because (a) there was yet no evidence base to roll-out the intervention for delivery at this second recruitment site, and (b) because a single-site randomized trial would have necessitated withholding a standard-care intervention from patients randomized to the control group, which would have raised considerable ethical and logistical concerns [37]. Pragmatic trial designs [38], where data collection has to complement current clinical delivery, are increasingly being used.

The intervention clinical team in the East of England were members of a pilot psycho-oncology service funded by a combination of third sector and local NHS awards. The service was designed to deliver an integrated psychological care model to patients with gynaecological cancers attending a large regional teaching hospital for treatment. The hospital provides comprehensive cancer care to patients both in the county in which it is located, and to out of county patients as a tertiary referral centre. The service was funded to see patients both in outpatient and inpatient settings and was located on the hospital site within the hospital's department of psychological medicine.

This study was reviewed by the University of Chester Department of Psychology Ethics Committee and was formally approved by the North Wales Research Ethics Committee (REC-4) (Reference: 18/WA/0079; IRAS Project ID: 239518). Health Research Authority permissions were approved at each clinical site. Written informed consent was obtained from all participants to take part in the trial.

The intervention

The intervention is a group-delivered, low-intensity CBT-based course that is offered as part of standard care to people diagnosed with gynaecological cancer across hospitals in the East Anglia region of England. The content broadly includes: (i) psychoeducation about the emotional impact of cancer survivorship; (ii) values-based behavioural activation; (iii) thought challenging negative and problematic cognitions; (iv) coping with uncertainty; (v) narrative therapy-based approaches to supporting identity concerns; and (vi) psycho-education about sleep hygiene and fatigue. The content of the intervention was not altered by the study protocol as our aim was to evaluate this clinical service as it was designed; some iterative changes were made over time, but these did not change the fundamental therapeutic objectives of the course. The intervention group is co-facilitated by a PWP and a clinical nurse specialist from the medical oncology team. The course delivery is supervised by a clinical psychologist. On design of the service (and research study) the intervention was intended to be delivered over six weekly sessions, however, from group two onwards, the content was spread out to include an additional week with an eighth week added as a wrap-up and review with no new content.

For this trial, the intervention was compared to a treatment-as-usual control group. Participants in this group received no additional intervention other than their standard clinical care and a weekly telephone call with the research for data collection purposes.

7

Participants

Eligibility criteria and sample size

Participants were women over 16 years of age who had completed first-line treatment following a gynaecological cancer diagnosis. We excluded those expected not to survive for the full five-month duration of the study (to mitigate against distress), those not fluent in English, and those unable to give informed consent.

Our sample size calculation was based on our original study aims of comparing between-group intervention effectiveness. To adequately power a two-arm experimental comparison (1- β = .90; α = .95; nine time-points) would have required ten participants per group to detect a medium interaction effect size *f* = .25 (medium effect size are indicated by NICE [39] as being necessary for clinical significance in the case of depression-related outcomes), though given that medium effect sizes are uncommon in psychosocial oncology intervention research we planned to recruit in excess of this.

Recruitment

At our intervention-delivery site, patients either self-referred or were informed about the availability of the intervention following completion of first line-treatment by their clinicians using an information sheet. There was no distress screening: all patients accessing the service were offered the chance to join the intervention if they wanted to. Those identified as highly distressed were, however, offered additional one-to-one psychological support as part of stepped-care service provision [40]. On setting up the service, the expectation was that four or five intervention groups would run each year, each with up to 15 participants per group. We expected that rates of consent for data to be used in our study would be relatively high, and therefore that we would be able to recruit 35-40 participants in a twelve-month period.

At the control sites (two general hospitals in North Wales), eligible participants were identified by clinical teams as part of routine weekly multidisciplinary team meetings. They were then invited to participate by their oncologist or specialist nurse. For this group we included patients prospectively diagnosed through the recruitment period, but also any retrospectively identified who had completed first-line treatment within a period of four months from the start of the study: this was comparable with the intervention group given that the intervention was planned to be delivered just four times per year, thus introducing delay between completing treatment and accessing the intervention. A recent review suggested that with clinician approach we could expect around 70-80% participation rates [41], though this is likely to reduce over time due to participant attrition. Our intention was to recruit 100 participants to this group (a) to provide a more representative

description of unmet needs in this group, and (b) to enable extraction of a sub-set of participants with similar distress profiles at baseline to the intervention group for matched analysis. For various logistic reasons we were restricted to recruiting our control sample for just a six-month period.

Data collection and measures

All participants completed self-reported outcome assessments at baseline (pre-intervention), and then weekly for seven subsequent weeks: this was intended to map onto data collection at each week of the intervention delivery, plus a one-week follow-up. From the second delivery iteration of the CBT group, however, an extra two weeks were added to the CBT programme. Given that (a) the majority of our control group data collection was already collected by this point, and (b) the new 'week eight session' was a review and wrap-up session only, we did not consider it hugely problematic that this was delivered after our final data collection point: our post-intervention data collection point thus remained at Week 7, which in essence became the final week of delivery in the review programme. The three-month follow-up data was collected via telephone interview conducted by the intervention delivery team. All data for control-arm participants was collected via telephone-based structured interviews at identical time-points to the intervention arm. Patients reported on their own socio-demographic profile. Basic diagnosis and treatment information was provided by the clinical teams, with participant consent.

Primary outcomes

Depression. The Patient Health Questionnaire (PHQ-9) [42] is a nine-item tool which uses a fourpoint rating scale (ranging from 0 to 3) to ask how often in the last two weeks participants have experienced symptoms pertaining to appetite, concentration, energy hopelessness, and suicidality. Higher scores indicate higher levels of depression. The PHQ-9 has been established as a valid and reliable measure of depression severity [42], and recent work has established it as a sensitive assessment tool for cancer-related distress [43].

Anxiety. The Generalized Anxiety Disorder questionnaire (GAD-7) [44] is a seven-item screening tool to how often in the last two weeks participants have experienced symptoms pertaining to feeling anxious, worried, had difficulty relaxing, and irritability. The GAD-7 uses a four-point rating scale (ranging from 0 to 3) and higher scores indicate higher anxiety. The GAD-7 has been established as an accurate measure of [45], including specifically within psychosocial oncology research [46].

Both the GAD-7 and the PHQ-9 were included as they form part of the minimum data set required for the evaluation of IAPT services in England.

Secondary outcomes

Quality of Life. The Functional Assessment of Cancer Therapy-General (FACT-G) is a 33-item tool commonly used to assess quality-of-life in patients receiving cancer treatment [47]. The FACT-G uses a five-point rating scale (ranging from 0 to 4) asking how often in the last seven days difficulties have been experienced across four dimensions of wellbeing: physical; social/family; emotional; functional. Higher scores indicate lower quality-of-life. The FACT-G is a well validated tool for measuring quality-of-life in cancer sample [48].

Unmet needs. The Short-Form Supportive Care Needs Survey (SCNS SF-34) [49] is a 34-item survey to assess level of unmet needs across five separate domains: psychological needs; sexuality needs; health system and information needs; physical daily needs; and, care and support needs. The SCNS SF-34 uses a five-point rating scale (ranging from 1 to 5) asking how often in the last month patients have required help across each of these unmet needs items.

Therapeutic process measures

To evaluate potential intervention mediators, we assessed variables that we expected might be changed by a psychological treatment protocol. To reduce participant burden these were assessed at baseline, post-intervention and follow-up only.

Thought intrusion. The Impact of Events Scale-Revised (IES-R) [50] is a 22-item measure of thought intrusion following a traumatic event. The IES-R uses a five-point rating scale (ranging from 0 to 4) asking how often in the past week patients have experienced stress reactions after traumatic life events. This is a valid and reliable measure of stress responses in different populations [51]. We used the thought intrusion sub-scale of the IES-R to measure whether the intervention has reduced the occurrence of problematic cognitions, as would be expected in CBT [52].

Self-efficacy. Psycho-educational component of the intervention were expected to increase participants' perceptions of their ability to self-manage the long-term consequences of their cancer and treatment. We assessed this using an 11-item cancer specific self-management self-efficacy scale [53,54] which is an adaptation of the Self-Efficacy for Managing Chronic Disease Scale [55], a widely used and well-validated measure. Each item requests a response using a 10-point scale (ranging from 1, not at all confident, to 10, totally confident). The adapted scale for cancer survivors has good reliability and psychometric properties [53].

Psychological flexibility. Given that the intervention content drew upon themes from third-wave interventions in addition to traditional CBT, we assessed sub-components of psychological flexibility (openness to experience, behavioural awareness, valued action) using the Comprehensive

Assessment of Acceptance and Commitment Therapy processes (CompACT) [56]. The CompACT is a 23-item measure which demonstrates good internal consistency across all three scoring clusters [56]; high scores indicate greater psychological *in*flexibility.

Statistical analyses

Analysis was undertaken using SPSS v24. All data were cleaned, recoded and reverse-scored as needed using standard scoring procedures. Where there was less than 10% missing data on self-report variables, mean-score imputation was used. Standard checks for data normality (box plot visual graphs and Shapiro Wilk test of sampling distributions), and sample size determinations, were used to determine whether parametric or non-parametric tests should be used.

Information on recruitment, response and attrition are reported descriptively so as to provide a narrative on the success (or lack thereof) of the study design. Baseline differences in these variables between the intervention and control group were explored using Mann Whitney U tests.

Our sample size was small, and our data were underpowered, but we nonetheless explored some preliminary analyses of our stated study aims. Our primary aim — intervention effectiveness on depression and anxiety — was addressed using 2 by 2 mixed ANOVAs (exploring changes from baseline to post-intervention between condition) and 2 by 3 mixed ANOVAs (exploring changes from baseline to post-intervention *and* follow-up between conditions). Secondary outcomes were also analysed using ANOVAs. Mediation analyses of therapeutic process variables on intervention outcomes was not possible due to limited sample sizes. We therefore used ANOVAs to explore differences between conditions and over-time in these variables instead.

Results

After extending our recruitment period to last almost two years (Recruitment: May 2018 to February 2020; Follow-up period closed in May 2020), we closed recruitment having fallen considerably short of our recruitment targets (see Figure 1). Sign-up for the intervention group was far lower than had been expected, and so fewer intervention groups were run. In total, just eight patients had taken part in one of three iterations of the CBT group. Of these eight, six consented for their data to be used in the study representing a 75% response rate. At our control site, just 26 participants consented to take part; this was a result of both (a) staff sickness at one site leading a period of non-recruitment, and (b) actual diagnosis rates being far lower than had originally been estimated by the local clinical teams. Over our recruitment period, 51 new patients were diagnosed; 36 of these met eligibility criteria (71% eligibility rate), and of those 26 consented for the study (72% consent rate).

Although this fell short of our recruitment target (for both groups) it was not possible to keep recruitment open indefinitely; funding to support data collection in North Wales had run out, and due to service re-organisation the group-based intervention in East Anglia was no longer going to be offered as standard care. We thus made the difficult decision to close the study short of target and disseminate our lessons learned (which this paper aims to do). In this context, we decided to conduct a more basic comparison of groups rather than adopting a matched-sample comparison as had been originally planned.

ADD FIGURE 1 ABOUT HERE

In the CBT group, only one participant completed all sessions and the follow-up. Two participants had sporadic completion of the intervention across all weeks; another participant did not attend Week 5; another missed Week 2 and did not complete follow-up; and two participants missed Weeks 3 and 4, but then resumed participation thereafter. One participant dropped out entirely from Week 4 onwards (overall attrition to follow-up = 16%; see Figure 1).

In the control group, six participants did not complete Week 2; four did not complete Week 3; three did not complete Week 4; three did not complete Week 5; six did not complete Week 6; and eight did not complete follow-up. No participant who completed baseline officially dropped out of the study (overall attrition to follow-up = 35%).

Sample description

Table 1 summarises key demographic and clinical characteristics of participants. There are some notable differences between the two groups; specifically, the control group were older and therefore more likely to be retired. The majority of our sample (both groups) were white and identified as heterosexual. The control group reported having received less support from local psychology or Macmillan services but this is perhaps because intervention group participants were informed about the study by the dedicated Macmillan-funded service that they were already engaged with. The sample was dominated by participants with ovarian cancer but this did not differ between groups and is representative of the broader gynaecological cancer population. The majority of participants had been diagnosed with Stage 3 cancer, though there was variation in diagnostic staging in both groups. All participants in the CBT group had received treatment with curative intent, whereas some participants in the control group had been treated with palliative intent. The majority of participants had been treated with either surgery alone, or surgery plus chemotherapy.

INSERT TABLE 1 ABOUT HERE

Primary outcome: depression and anxiety

Participants in the CBT group had higher mean scores on both depression and anxiety than the control group at baseline, with the difference in anxiety reaching statistical significance (p=.03; see Table 2). Scores decreased to both post-intervention and follow-up in both groups, and we note they are higher in control group at post-treatment compared to those who received the intervention. As an assumption check for ANOVA tests, Levene's tests for homogeneity of variance (taken from the 2*3 ANOVAs) demonstrated that the assumption of equal variances between groups was not violated for either depression (baseline, p = .157; post-treatment, p = .449; follow-up, p = .907) or anxiety (baseline, p = .844; post-treatment, p = .440; follow-up, p = .605).

INSERT TABLE 2 ABOUT HERE

For depression, comparing only baseline to post-intervention changes, we found a significant main effect of time (F[1, 27] = 13.43, p = .001, $n^2_{par} = .33$), but a non-significant main effect of condition (F[1,27] = .63, p = .435, $n^2_{par} = .02$) and a non-significant time by condition interaction effect (F[1, 27] = 2.44, p = .130, $n^2_{par} = .08$). Using a 3*2 ANOVA which also incorporated follow-up scores, neither the main effect of time (F[2, 34] = 1.99, p = .152, $n^2_{par} = .11$), or condition (F[1,17] = .60, p = .45, $n^2_{par} = .03$), nor a time by condition interaction (F[2, 34] = .03, p = .973, $n^2_{par} = .00$) reached significance.

Similarly, for anxiety, comparing baseline to post-intervention scores, our data show a significant main effect of time (F[1, 25] = 15.854, p = .001, $n_{par}^2 = .39$), but non-significant main effects of condition (F[1, 24] = .409, p = .528, $n_{par}^2 = .02$) and time by condition interaction (F[1, 25] = 5.84, p = .023, $n_{par}^2 = .19$). Extending this to include follow-up data (comparing baseline, post and follow-up) resulted in the main effect of time remaining significant (F[2, 34] = 11.07, p < .001, $n_{par}^2 = .39$) and additional significant differences emerging for main effects of condition (F[1, 17] = .21.12, p < .001, $n_{par}^2 = .56$) and time by condition interaction (F[2, 34] = 3.51, p = .04, $n_{par}^2 = .17$).

Secondary outcomes: quality of life and unmet needs

Participants in the CBT group had lower mean scores on quality of life at baseline and at postintervention than those in the control condition. Mean scores were roughly equivalent at follow-up for both groups. Levene's tests for homogeneity of variance demonstrated that the equal variances assumption was violated for quality of life at baseline (p = .034) and follow-up (p = .044) (2*2 model only).

Baseline to post-intervention change in quality of life failed to reach significance for main effects of time (F[1, 27] = .28, p = .600, n_{par}^2 = .01), condition (F[1, 7] = .21, p = .635, n_{par}^2 = .01) or time by condition interaction (F[1, 27] = .04, p = .847, n_{par}^2 = .00). Including follow-up data failed to improve

these findings: no main effects of time (F[2, 24] = .29, p = .749, $n^2_{par} = .02$), condition (F[1, 12] = .60, p = .453, $n^2_{par} = .05$), or time by condition interaction (F[2, 24] = .78, p = .472, $n^2_{par} = .06$) effects reached significant. However, these latter results should be interpreted with caution given that there were only 2 participants in the CBT group and 12 participants in the control group.

Moderate to high needs were defined as a score of four or five on each item of the SCNS-SF34, with a score less than three indicating no need at present. Regardless of group allocation, all participants had at least one moderate-to-high need in the patient care and support needs domain. The highest rated need was *'Reassurance by medical staff that the way you feel is normal'* for the CBT group participants, and *'Hospital staff attending promptly to your physical needs'* for the control group. Given the small sample size and some violations of normality on these variables, we were unable to run parametric statistical analysis; instead, Mann-Whitney U test confirmed that there were no differences between group on overall domain scores of psychological needs, sexuality needs, health system and information needs, physical daily needs or care and support needs (Table 2). Unmet needs reduced over time for both groups, and in all domains. A 2*2 ANOVA on total needs score demonstrated a significant main effect of time (*F*[1, 19] = .07, *p* = .799, n²_{par} = .00) and time by condition interaction effect (*F*[1, 19] = .11, *p* = .109, n²_{par} = .01).

Intervention mediator variables

Due to our limited sample size, we were unable to undertake statistical analysis of potential therapeutic process as outcome mediator variables. At baseline, thought intrusion was higher in the CBT group and cancer-specific self-management self-efficacy was higher in the control group. At baseline, the CBT group also scored higher in overall psychological flexibility and sub-scale scores of behavioural activation and value-action; interestingly, however, our control group scored as more inflexible on the openness to experience subscale. By follow-up, between group differences had minimised with the CBT group reducing in intrusive thoughts, increasing in self-management self-efficacy and reduced psychological inflexibility, though there was high variance on each, as indicated by large standard deviations.

Two-by-two mixed ANOVAs were run to explore between group differences in score changes from baseline to follow-up (Table 3). These demonstrate no significant main effects or interaction effects for thought intrusion. A significant main effect of time was found for self-management self-efficacy, and whilst the main effect of condition failed to reach significance, the time by condition interaction was significant. For psychological inflexibility (total score), there was only a significant main effect of time only.

INSERT TABLE 3 ABOUT HERE

Discussion

With the potential for the benefit of psychological intervention on distress and quality of life in cancer so readily acknowledged [13, 57], research that expands the evidence both in terms of efficacy *and* readily implementable interventions is needed. Given the importance of cost-effectiveness [24], group-based, low-intensity interventions are attractive, and the growing role of Psychological Wellbeing Practitioners (PWPs) in managing the psychological effects of chronic health conditions in the UK offers an obvious route through which to test the feasibility of these types of interventions. The study, therefore, aimed to explore the effectiveness of a PWP co-facilitated, group-based, psychological intervention for improving depression and anxiety in women treated for gynaecological cancer. As secondary aims, we wanted to explore intervention effects on quality of life and unmet needs, and to identify potential psychotherapeutic mediators of intervention outcomes.

Effectiveness of the intervention

A series of practical issues resulted in low recruitment and so our data was not sufficiently powered to address any of these aims conclusively. Nonetheless, there are some interesting patterns in the data. Baseline scores of anxiety and depression were higher in the intervention group; this is not surprising given that they had presented for additional psychological support, compared to our control group which was a population snapshot. Of importance, this probably reinforces the idea that not all cancer survivors suffer with problematic levels of psychological distress, and even in those that do, only a small minority will want this to be addressed with a formal psychology-based [58]. Depression improved over time in all participants however, no between-group differences were identified. Anxiety, in comparison, showed more intervention benefits with significant between-group differences emerging from baseline to follow-up. It is possible that the lack of statistical power is masking the true effectiveness of the intervention, but as they stand, these data suggest that the intervention might be more effective for anxiety outcomes than it is on targeting improvements in depression. Such a difference might not be surprising given the different causal factors that may be at play in the development of these distress components in cancer [59], and thus the different intervention strategies that might be needed.

The most frequently endorsed unmet needs in our sample were those in the 'Patient Care and Support' domain. This is in contrast to other studies that suggests that psychological and sexuality needs are more prominent in this survivor group [10,11]. Unmet needs reduced over time in both

groups as might be expected as part of the natural adaptation process at the end of treatment, but we identified no additional intervention effects on this speed or extent of this improvement.

There is a paucity of research investigating potential psychological mediators in cancer survivors [15], therefore the EPELIT study was designed to also explore potential mediators. This is important: where the most effective mediators can be identified, intervention content can be refined to maximise change on these process variables, to consequently maximise improvements in desired outcome variables. We urge caution interpreting the results of changes in putative psychotherapeutic mediators from our study, however we are encouraged by the significant time by condition interaction for self-management self-efficacy. Given the literature elsewhere on selfmanagement self-efficacy as a useful coping and adaptive skill [54], it is possible that this might have accounted for some of the improvement in anxiety in our sample. Further work to test this in larger, more powered samples, is now needed.

Although we did not collect data on intervention acceptability, this is clearly of interest given the novel facilitation method of this intervention, such as the use of a PWP intervention facilitation. Drop-out from psychological interventions can be a useful proxy indicator of acceptability and to this end, the data from this study are encouraging and suggest that CBT remains an acceptable intervention framework for this group [15], and that delivery by a team including PWPs was not off-putting to those who agreed to participate.

Study evaluation and methodological discussion

Though our participant numbers were low, response rates were encouraging. Because study invitation was provided by a known member of the clinical care team, we had expected around a 70-80% participation rate [41], and our study data bear this out: we achieved 72 and 75% participation rates in the control and intervention group respectively. Retention to follow-up was also good with attrition of just 16% in the intervention group and a higher, but still non-problematic, 35% in the control group. Our failure to successfully recruit was not, therefore, a problem of consent or retention, but rather one of baseline population size: at our control site, the number of new diagnosis was just one third of the number that we had been led to expect on designing the trial, and at the intervention site far fewer participants were referred (or self-referred) to this intervention than had been expected, falling short of expectations by a factor of four. To further clarify, we even collected data on the reasons that patients at each site might not have been recruited. These data show that this was most definitely not an issue of high levels of ineligibility: all those presenting for the intervention group were eligible for the study and 71% of all new patients diagnosed at the control site were eligible and approached. There were some demographic and

clinical biases in our sample, but these are representative of the clinical and demographic population from which we recruited. Nonetheless, efforts to recruit a more demographically diverse sample in future research would be welcomed as a way to expand the overall evidence-base in psychosocial oncology [15].

In this way, perhaps paradoxically, the greatest strength of our study design was also the greatest weakness. We selected a pragmatic non-randomised trial design for this study with both practical and ethical considerations in mind; these kinds of design are becoming more and more popular in clinical care research because they provide a way for existing services to be evaluated against a control group, without the ethical concerns of having to withhold a potentially beneficial intervention to which all patients *should* have access from the control group. It was, however, this design which led to the ultimate failure of the study: existing services, by their very design, have to be responsive to the populations that they serve, are subject to changes in staffing, funding and commissioning, and so iterative changes are to be expected over time. In our case, some of these changes introduced unexpected methodological challenges and potential bias.

We became aware that some of the intervention content and order was modified followed a change in staffing part-way through the study. Though this was not intended to change any of the fundamental psychotherapeutic goals of the intervention, changing content in this way may potentially have changed the macro-level focus of different underlying psychotherapeutic process. Had we recruited to target this would have undoubtedly made analysis of mediator data incredibly complex, if not entirely uninterpretable. Again, whilst understandable from a service-delivery perspective, this can have profound consequences on the methodological validity of aligned effectiveness trials.

Relatedly, with our post-intervention data point scheduled for week seven (one week after the close of the original design of the group intervention) this change to the intervention programme meant that for those participating in the study after this change, the post-intervention questionnaire assessment was completed after receiving only 75% of the intervention sessions. Our trial steering group made the decision not to change the timing of this data collection point given that a considerable number of control-group participants had already been recruited and this would have rendered their data non-comparable. Given variance in exact completion time of longer-term followup across the literature, we do not anticipate that this change would have affected our final threemonth follow-up data, but it may well have affected earlier time-points. This highlights the challenges of evaluating existing services which have to be responsive to local clinical need, and the potential benefits of testing these kinds of research questions using an 'additive' model where more research control is retained over intervention content and delivery.

17

As a final important critical note, we would add that psychosocial research is often limited because the outcomes to which we are most interested are burdensome to measure. Unlike disease-oriented outcomes such as number of participants who experienced disease recurrence or survival, outcomes such as distress and quality of life are subjective, and thus require assessment using patientreported measures. These measurement tools can feel burdensome and so we often attempt to minimise the length and number of administrations when designing studies; indeed, this is a common concern of ethical review committees. To investigate intervention mediators effectively, however, regular assessments are important; an important secondary research question in our study, therefore, was to establish the willingness of participants to report on these variables on a weekly basis. To this end, our study was a partial success: though we report 35% follow-up attrition, our discussions with participants indicated that a vast majority of these missing data points were about scheduling difficulties more than they were about participation fatigue. This is an important learning point from our work which demonstrates that we should not necessarily be quite so avoidant of higher-burden data collection protocols in our trial research.

Implications and future research

Some implications from this study are somewhat obvious: because of recruitment issues, our study failed to answer conclusively whether (a) CBT-based interventions are helpful in improving psychological wellbeing and (b) whether psychological wellbeing practitioner facilitation is an effective delivery mechanism. Future research to address these important research questions is still very much needed, but a more traditional approach using a randomised controlled trial *independent of usual care* might be a more effective study design.

Of perhaps greater concern: whilst smaller than in some comparison cancer groups, increasing survivorship within gynaecological cancer means that far more women are living after treatment for this kind of cancer over time [2] and previous research demonstrates that unmet psychological and physical needs often remain high [60]. Recruitment into these intervention groups should, therefore, have been simple. Our study failed to recruit, not because women attending the groups did not consent to be part of the trial, but simply because the intervention groups themselves failed to recruit. Work is needed, therefore, to understand why those with high needs are not presenting for this kind of psychological support: qualitative studies on the barriers to attendance or why this particular offering was unappealing would be helpful. Indeed, the future design of such interventions might benefit from using a co-design method of production [61] to ensure that what is offered is both acceptable to, and wanted by, those for who the services are created.

Conclusions

This trial did not go to plan: although indications are that the intervention was acceptable and potentially helpful, recruitment was incredibly challenging and the iterative nature of development of the intervention rendered concrete conclusions impossible. Whilst we have demonstrated the success of some of our methodological choices (e.g. frequent data collection for designs intended to explore mediation), our overall design choice of employing a pragmatic trial design was not appropriate because of the lack of control we had over both recruitment and changes to intervention content. This has important implications for future research in psychosocial oncology; whilst pragmatic trials might be appropriate for later-stage implementation research, we found them to be fraught with methodological challenges for earlier-stage effectiveness trials and would urge others to use them with caution.

Data availability

Underlying data

Due to the information provided to the approving ethics committee and to individual participants at the point of informed consent, the underlying data for this study cannot be placed on an open access repository. An anonymised data set is available on request from the corresponding author. This dataset will include self-report measures, demographic information as reported to the research team, and clinical data as extracted from patient medical records; it will not include information on hospital site to minimise the likelihood of any individual participant being identifiable. Provided that the data are intended for research use (including verification of the analysis reported in this paper or for the exploration of additional research questions), requests will not be unnecessarily declined.

Reporting guidelines

Repository name: CONSORT checklist for 'Evaluating process and effectiveness of a low-intensity CBT intervention for women with gynaecological cancer (the EPELIT Trial)', DOI 10.17605/OSF.IO/YGHNE, Released under a CC0 Creative Commons License.

Acknowledgements

The authors would like to thank Emma Offord for her support in planning the study, and to Asmita Dhungana and Kathrin Hicks for supporting data collection and recruitment at the intervention site. Thanks to Sharon Manning and Caryl Butterworth for supporting recruitment in the control site.

Conflicts of Interest

The authors declare no conflicts of interest.

Funding

This study was funded by a QR (Psychology) Grant from the University of Chester. The clinical service evaluated as part of this study was funded by Macmillan Cancer Support.

References

- [1] Office for National Statistics. Cancer registration statistics, England: 2017. Downloaded from https://www.ons.gov.uk/peoplepopulationandcommunity/healthandsocialcare/conditionsan ddiseases/bulletins/cancerregistrationstatisticsengland/latest.
- [2] Youlden D, Coory M & Baade PD. Cancer Survival in Queensland 2002. Brisbane, Queensland Health and Queensland Fund; 2005. Downloaded from https://cancerqld.azurewebsites.net/wpcontent/uploads/2015/11/60144_cancer_survival_in_queensland_2002.pdf.
- [3] Coleman MP, Forman D, Bryant H ... & Richards MA. Cancer survival in Australia, Canada, Denmark, Norway, Sweden, and the UK, 1995-2007 (the International Cancer Benchmarking Partnership): an analysis of population-based cancer registry data. *Lancet* 2011; 377 (9760): 127-138. DOI: 10.1016/S0140-6736(10)62231-3
- [4] Neal RD, Din NU, Hamilton W ... & Rubin G. Comparison of cancer diagnostic intervals before and after implementation of NICE guidelines: analysis of data from the UK General Practice Research Database. *Br J Cancer* 2014; *110*(3): 584-592. DOI:10.1038/bjc.2013.791.
- [5] Macleod U, Mitchell ED, Burgess C, Macdonal S & Ramirez AJ. Risk factors for delayed presentation and referral of symptomatic cancer: evidence for common cancers. *Br J Cancer* 2009; 101: S92-S101. DOI: 10.1038/sj.bjc.6605398
- [6] Risberg T, Sørbye SW, Norum J & Wist EA. Diagnostic delay causes more psychological distress in female than in male cancer patients. *Anticancer Res* 1996; 16(2): 995-999.
- [7] Batty GD, Russ TC, Stamatakis E & Kivimäki M. Psychological distress in relation to site specific cancer mortality: pooling of unpublished data from 16 prospective cohort studies. *BMJ* 2017; 356: j108. DOI:10.1136/bmj.j108.

- [8] Ashing-Giwa KT, Tejero JS, Kim J ...& Lim JW. Cervical cancer survivorship in a population based sample. *Gynecol Oncol* 2009; 112(2): 358-364. DOI:10.1016/j.ygyno.2008.11.002.
- Urbaniec OA, Collins K, Denson LA & Whitford HS. Gynecological cancer survivors: assessment of psychological distress and unmet supportive care needs. *J Psychosoc Oncol* 2011; 29(5): 534-551. DOI:10.1080/07347332.2011.599829.
- [10] Reb AM & Cope DG. Quality of life and supportive care needs of gynecologic cancer survivors. West J Nurs Res 2019; 41(10): 1385-1406. DOI: 10.1177/0193945919846901.
- [11] Albers LF, van Belzen MA, van Batenburg C ... & Elzevier HW. Discussing sexuality in cancer care: towards personalized information for cancer patients and survivors. Sup Care Cancer, 2020; 28(9): 4227-4233. DOI: 10.1007/s00520-019-05257-3.
- [12] Beesley V, Eakin E, Steginga S, Aitken J, Dunn J & Battistutta D. Unmet needs of gynaecological cancer survivors: implications for developing community support services. *PsychoOncol 2008;* 17(4): 392-400. DOI:10.1002/pon.1249.
- [13] Ahmed-Lecheheb D & Foly F. Ovarian cancer survivors' quality of life: a systematic review. J Cancer Surviv 2016; 10(5): 789-801. DOI: 10.1007/s11764-016-0525-8.
- [14] Faller H, Schuler M, Richard M, Heckl U, Weis J & Küffner R. Effects of psycho-oncologic interventions on emotional distress and quality of life in adult patients with cancer: systematic review and meta-analysis. *J Clin Oncol* 2013; 31(6): 782-793.
 DOI:10.1200/JCO.2011.40.8922.
- [15] Hulbert-Williams NJ, Beatty L & Dhillon HM. Psychological support for patients with cancer:
 evidence review and suggestions for future directions. *Curr Opin Support Palliat Care;* 2018;
 12(3): 276-292. DOI: 10.1097/SPC.00000000000360.
- [16] Matthews H, Grunfeld EA & Turner A. The efficacy of interventions to improve psychosocial outcomes following surgical treatment for breast cancer: a systematic review and metaanalysis. *PsychoOncol* 2017; 26(5): 593-607. DOI:10.1002/pon.4199.
- [17] Xiao F, Song X, Chen Q ... & Guo Q. Effectiveness of psychological interventions on depression in patients after breast cancer surgery: a meta-analysis of randomized controlled trials. *Clin Br Cancer* 2017; 17(3): 171-179. DOI: 10.1016/j.clbc.2016.11.003.

- [18] Beesley VL, Alemayehu C & Webb PM. A systematic literature review of trials of survivorship interventions for women with gynaecological cancer and their caregivers. *Eur J Cancer Care*, 2019; 28(3): e13057. DOI: 10.1111/ecc.13057/.
- [19] Rost AD, Wilson K, Buchanan E, Hildebrandt MJ & Mutch D. Improving psychological adjustment among late-stage ovarian cancer patients: examining the role of avoidance in treatment. *Cog Behav Prac* 2012; 19(4): 508-517. DOI:10.1016/j.cbpra.2012.01.003.
- [20] Donnelly CM, Blaney JM, Lowe-Strong ... & Gracey JH. A randomised controlled trial testing the feasibility and efficacy of a physical activity behavioural change intervention in managing fatigue with gynaecological cancer survivors. *Gynecol Oncol* 2011; *122*(3): 618-624. DOI:10.1016/j.ygyno.2011.05.029.
- [21] Moonsammy SH, Guglietti CL, Santa Mina D ... & Ritvo P. A pilot study of an exercise & cognitive behavioral therapy intervention for epithelial ovarian cancer patients. J Ovarian Res 2013; 6(1): 21. DOI:10.1186/1757-2215-6-21.
- [22] Brotto LA, Erskine Y, Carey M, ... & Miller D. A brief mindfulness-based cognitive behavioral intervention improves sexual functioning versus wait-list control in women treated for gynecologic cancer. *Gynecol Oncol* 2012; 125(2): 320-325. DOI:10.1016/j.ygyno.2012.01.035.
- [23] Chan YM, Lee PW, Fong DY, ... & Wong LC. Effect of individual psychological intervention in Chinese women with gynecologic malignancy: a randomized controlled trial. J Clin Oncol 2015; 23(22): 4913-4924. doi:10.1200/JCO.2005.02.069.
- [24] Owen JE, Klapow JC, Hicken B & Tucker DC. Psychosocial interventions for cancer:
 review and analysis using a three-tiered outcomes model. *PsychoOncol 2001;* 10(3): 218-30. DOI:10.1002/pon.509.
- [25] Bower P & Gilbody S. Stepped care in psychological therapies: access, effectiveness and efficiency: narrative literature review. *Brit J Psychiatry*, 2005; 186(1): 11-17.
 DOI:10.1192/bjp.186.1.11.
- [26] Lambert M. Presidential address: What we have learned from a decade of research aimed at improving psychotherapy outcome in routine care. *Psychother Res* 2007; 17(1): 1-14. DOI:10.1080/10503300601032506.

- [27] CSIP Choice and Access Team. Improving Access to Psychological Therapies (IAPT) Commissioning Toolkit. London: Department of Health, 2008. Downloaded from https://www.england.nhs.uk/mental-health/adults/iapt/.
- [28] Green H, Barkham M, Kellett S & Saxon D. Therapist Effects and IAPT Psychological Wellbeing Practitioners (PWPs): A multilevel modelling and mixed methods analysis. *Behav Res Ther*, 2014; 63: 43-54. DOI: 10.1016/j.brat.2014.08.009.
- [29] Bennett-Levy J, Richards DA, Farrand P ... & Proudfoot J. Low intensity CBT interventions: a revolution in mental health care. In Bennett-Levy DA, Richards P, Farrand H et al. (Eds.), Oxford guide to low intensity CBT interventions, 2010; 3-18.
- [30] NHS England. Implementing the five year forward view for mental health. London: NHS England, 2016. Downloaded from https://www.england.nhs.uk/mentalhealth/taskforce/imp/.
- [31] Edelman S, Bell DR & Kidman AD. A group cognitive behaviour therapy programme with metastatic breast cancer patients. *PsychoOncol* 1999; 8(4): 295-305. DOI:10.1002/(SICI)1099-1611(199907/08)8:4<295::AID-PON386>3.0.CO;2-Y.
- [32] Savard J, Simard S, Ivers H & Morin CM. Randomized study on the efficacy of cognitivebehavioral therapy for insomnia secondary to breast cancer, part I: Sleep and psychological effects. *J Clin Oncol* 2005; 23(25): 6083-6096.
 DOI:10.1200/JCO.2005.09.548.
- [33] Beatty L, Kemp E, Butow P & Koczwara B. A systematic review of psychotherapeutic interventions for women with metastatic breast cancer: Context matters. *PsychoOncology*, 2017; 27(1): 34-42. DOI: 10.1002/pon.4445.
- [34] Masheb RM, Kerns RD, Lozano C, Minkin MJ & Richman S. A randomized clinical trial for women with vulvodynia: Cognitive-behavioral therapy vs. supportive psychotherapy. *Pain* 2009; 141(1-2): 31-40. DOI:10.1016/j.pain.2008.09.031.
- [35] Sekse RJT, Blaaka G, Buestad I, Tengesdal E, Paulsen A & Vika M. Education and counselling group intervention for women treated for gynaecological cancer: does it help?. Scandin J Caring Sci 2004; 28(1): 112-121. DOI:10.1111/scs.12024.

- [36] Lazenby M, Tan H, Pasacreta N, Ercolano E & McCorkle R. The five steps of comprehensive psychosocial distress screening. *Curr Oncol Rep* 2015; 17(5): 22. DOI:10.1007/s11912-015-0447-z.
- [37] Hart T Fan JR & Novack TA. The dilemma of the control conditions in experience-based cognitive and behavioural treatment research. *Neuropsych Rehab* 2008; 18(1): 1-21. DOI: 10.1080/09602010601082359.
- [38] Ford I & Norrie J. Pragmatic Trials. N Engl J Med 2016; 375(5): 454-463.
 DOI:10.1056/NEJMra1510059.
- [39] NICE. National Clinical Practice Guideline No 23, (October 2003), 33–46. Downloaded from https://www.nice.org.uk/guidance/cg90/evidence/cg90-depression-in-adults- fullguidance appendix-212.
- [40] Hutchison SD, Steginga SK & Dunn J. The tiered model of psychosocial intervention in cancer: a community based approach. *PsychoOncol* 2005; 15(6): 541-546. DOI:10.1002/pon.973.
- [41] Wakefield CE, Fardell JE, Doolan EL ... & King, M. Participation in psychosocial oncology and quality-of-life research: a systematic review. *Lancet Oncol* 2017; 18(3): e153-e165.
 DOI:10.1016/S1470-2045(17)30100-6.
- [42] Kroenke K, Spitzer RL & Williams JB. The PHQ-9. J Gener Intern Med 2009; 16(9): 606-613.
 DOI:10.1046/j.1525-1497.2001.016009606.x.
- [43] Hartung TJ, Friedrich M, Johansen C ... & Wegscheider K. The Hospital Anxiety and Depression Scale (HADS) and the 9-item Patient Health Questionnaire (PHQ-9) as screening instruments for depression in patients with cancer. *Cancer* 2017; 123(21): 4236-4243.
 DOI:10.1002/cncr.30846.
- [44] Spitzer RL, Kroenke K, Williams JB & Löwe B. A brief measure for assessing generalized anxiety disorder: the GAD-7. *Arch Intern Med* 2006; 166(10): 1092-1097.
 DOI:10.1001/archinte.166.10.1092.
- [45] Swinson RP. The GAD-7 scale was accurate for diagnosing generalised anxiety disorder. BMJ
 Evid Based Med 2006; 11(6): 184. DOI:10.1136/ebm.11.6.184.

- [46] Esser P, Hartung TJ, Friedrich M... & Wegscheider K. The Generalized Anxiety Disorder
 Screener (GAD-7) and the anxiety module of the Hospital and Depression Scale (HADS-A) as
 screening tools for generalized anxiety disorder among cancer patients. *PsychoOncol* 2018;
 27(6): 1509-1516. DOI:10.1002/pon.4681.
- [47] Cella DF, Tulsky DS, Gray G ... & Brannon J. The Functional Assessment of Cancer Therapy scale: development and validation of the general measure. *J Clin Oncol* 1993; 11(3): 570-579.
 DOI:10.1200/JCO.1993.11.3.570.
- [48] Luckett T, King MT, Butow ... & Heading G. Choosing between the EORTC QLQ-C30 and FACT-G for measuring health-related quality of life in cancer clinical research: issues, evidence and recommendations. *Annal Oncol* 2011; 22(10): 2179-2190. DOI:10.1093/annonc/mdq721.
- [49] Boyes A, Girgis A & Lecathelinais C. Brief assessment of adult cancer patients' perceived needs: development and validation of the 34-item Supportive Care Needs Survey (SCNS-SF34). J Eval Clin Prac 2009; 15(4): 602-606. DOI:10.1111/j.1365-2753.2008.01057.x.
- [50] Weiss DS. The Impact of Event Scale: Revised. In Wilson JP & Tan CS (eds), Cross-cultural assessment of psychological trauma and PTSD (pp. 219-238). Springer, Boston, MA; 2007. DOI:10.1007/978-0-387-70990-1_10.
- [51] Sundin EC & Horowitz MJ. Horowitz's Impact of Event Scale evaluation of 20 years of use. *Psychosom Med* 2003; 65(5): 870-876. DOI:10.1097/01.PSY.0000084835.46074.F0.
- [52] Clark DA & Radomsky AS. Introduction: A global perspective on unwanted intrusive thoughts. *J Obsessive Compuls Relat Disord* 2014; 3(3): 265-268.
 DOI:10.1016/j.jocrd.2014.02.001.
- [53] Foster C, Breckons M, Hankins M, Fenlon D, Cotterell P. Developing a scale to measure self-efficacy to self-manage problems following cancer treatment. *Psycho* Oncol 2013; 22(S1): 16. DOI: 10.1111/j.1099-1611.2013.03239.x
- [54] Foster C, Breckons M, Cotterell P ... & Smith PW. Cancer survivors' self-efficacy to self-manage in the year following primary treatment. *J Cancer Surviv* 2015; 9(1): 11-19.
 DOI:10.1007/s11764-014-0384-0.

- [55] Self-Efficacy for Managing Chronic Disease Scale. Stanford Self-Management Resource Centre; 2013. Downloaded from https://www.selfmanagementresource.com/resources/evaluation-tools/
- [56] Francis AW, Dawson DL & Golijani-Moghaddam N. The development and validation of the Comprehensive assessment of Acceptance and Commitment Therapy processes (CompACT).
 J Context Behav Sci 2016; 5(3):134-145. DOI:10.1016/j.jcbs.2016.05.003.
- [57] Faller H, Brähler E, Härter ... & Sehner S. Unmet needs for information and psychosocial support in relation to quality of life and emotional distress. A comparison between gynecological and breast cancer patients. *Patient Educ Couns* 2017; 100(10): 1934-1942. DOI:10.1016/j.pec.2017.05.031.
- [58] Lambert SD, Kelly B, Boyes A & Girgis A. Insights into preference for psycho-oncology services among women with gynecologic cancer following distress screening. J Natl Compr Canc Netw 2014; 12(6): 899-906. DOI: 10.6004/jnccn.2014.0084.
- [59] Hulbert-Williams NJ, Neal RD, Morrison V, Hood K & Wilkinson C. Anxiety, depression and quality of life after cancer diagnosis: what psychosocial variables best predict how patients adjust? *PsychoOncol* 2012; 21(8): 857-867. DOI: 10.1002/pon.1980.
- [60] Simonelli LE & Pasipanodya E. Health disparities in unmet support needs of women with gynecologic cancer: an exploratory study. *J Psychosoc Oncol* 2014; 32(6): 727-734.
 DOI:10.1080/07347332.2014.955240.
- [61] Bate P & Robert G. Experience-based design: from redesigning the system around the patient to co-designing services with the patient. *BMJ Qual Saf* 2006; 15(5): 307-310. DOI: 10.1136/qshc.2005.016527.

| | · · · | |
|-----------------------|----------------------------|---------------------------------|
| | CBT (<i>n</i> = 6) | Control (<i>n</i> = 26) |
| Age M (SD) | 44.50 (13.07) | 68.81 (9.52) |
| Dependents M (SD) | .17 (.41) | .31 (.55) |
| Employment status | | |
| Unemployed | 1 (16.7%) | 2 (7.7%) |
| Employed part-time | 4 (66.7%) | 2 (7.7%) |
| Employed full-time | 0 | 1 (3.8%) |
| Self-employed | 0 | 2 (7.7%) |
| Retired | 1 (16.7%) | 18 (69.2%) |
| Other | 0 | 1 (3.8%) |
| Relationship status | | |
| Yes, living together | 5 (83.3%) | 19 (73.1%) |
| Yes, living apart | 0 | 3 (11.5%) |
| No | 1 (16.7%) | 4 (15.4%) |
| Ethnicity | | |
| White | 5 (83.3%) | 100% |
| Other | 1 (16.7%) | 0 |
| Sexuality | | |
| Heterosexual | 6 (100%) | 25 (96.2%) |
| Missing | 0 | 1 (3.8%) |
| Support received | | |
| Macmillan information | 4 (66.7%) | 11 (42.3%) |
| Psychologist | 1 (16.7%) | 0 |
| None | 0 | 8 (30.8%) |
| Other | 1 (16.7%) | 7 (26.9%) |
| Cancer type | | |
| Ovarian | 4 (66.7%) | 13 (50.0%) |
| Peritoneal | 0 | 1 (3.8%) |
| Endometrial | 1 (16.7%) | 6 (23.1%) |
| Cervix | 1 (16.7%) | 1 (3.8%) |
| Ovarian-Endometrial | 0 | 1 (3.8%) |
| | | |

Table 1. Description of demographic and clinical profile of participants.

| Vulval | 0 | 1 (3.8%) |
|---------|---|----------|
| Missing | 0 | 1 (3.8%) |

| Cancer stage | at diagnosis |
|--------------|--------------|
|--------------|--------------|

| 1a | 0 | 5 (19.2%) |
|-----------------------------|-----------|------------|
| 1b | 0 | 3 (11.5%) |
| 1c | 1 (16.7%) | 1 (3.8%) |
| 2a | 0 | 1 (3.8%) |
| 2b | 1 (16.7%) | 1 (3.8%) |
| За | 1 (16.7%) | 2 (7.7%) |
| 3c | 1 (16.7%) | 7 (26.9%) |
| 4a | 0 | 2 (7.7%) |
| 4b | 1 (16.7%) | 3 (11.5%) |
| Missing | 0 | 1 (3.8%) |
| Treatment intent | | |
| Palliative | 0 | 7 (26.9%) |
| Curative | 6 (100%) | 18 (69.2%) |
| Treatment received | | |
| Surgery only | 0 | 5 (19.2%) |
| Surgery + chemotherapy | 5 (83.3%) | 14 (53.8%) |
| Surgery + radiotherapy | 0 | 3 (11.54%) |
| Radiotherapy + chemotherapy | 1 (16.7%) | 0 |
| Missing | 0 | 1 (3.8%) |

| | СВТ | Control | Baseline differences |
|-------------------------------------|---------------|---------------|----------------------|
| Depression | | | |
| Baseline | 9.50 (3.99) | 5.58 (5.01) | U = 40.50, p = .069 |
| Post | 3.20 (2.68) | 3.58 (3.20) | |
| Follow-up | 4.00 (2.83) | 2.24 (3.13) | |
| Anxiety | | | |
| Baseline | 8.83 (5.31) | 4.08 (3.19) | U = 32.00, p = .030 |
| Post | 1.80 (1.30) | 2.78 (3.44) | |
| Follow-up | 2.67 (2.52) | 2.24 (3.13) | |
| Quality of life | | | |
| Baseline | 53.97 (4.19) | 54.90 (8.04) | U = 61.50, p = .425 |
| Post | 53.73 (3.54) | 55.96 (8.04) | |
| Follow-up | 59.00 (1.41) | 59.00 (7.01) | |
| Care needs total | | | |
| Baseline | 9.67 (3.32) | 10.23 (2.50) | U = 64.00, p = .499 |
| Follow-up | 7.50 (1.91) | 8.00 (1.77) | |
| Sexuality care needs | | | |
| Baseline | 7.67 (4.55) | 4.85 (2.17) | U = 50.50, p = .174 |
| Follow-up | 6.25 (4.27) | 3.82 (1.94) | |
| Physical daily living needs | | | |
| Baseline | 10.83 (1.72) | 11.12 (4.59) | U = 29.50, p = .056 |
| Follow-up | 8.50 (2.89) | 8.59 (4.37) | |
| Psychological needs | | | |
| Baseline | 28.43 (12.27) | 22.62 (7.25) | U = 54.50, p = .256 |
| Follow-up | 20.00 (8.60) | 18.12 (7.99) | |
| Health system needs | | | |
| Baseline | 21.00 (5.29) | 22.88 (7.25) | U = 69.50, p = .673 |
| Follow-up | 15.75 (3.95) | 19.88 (3.44) | |
| Psychological inflexibility (total) | | | |
| Baseline | 70.00 (29.35) | 45.46 (12.87) | U = 26.00, p = .012 |
| | | | |

Table 2. Means and standard deviations for each variable across groups and time.

| Follow-up | 39.25 (16.38) | 36.88 (12.85) | |
|----------------------|---------------|---------------|---------------------|
| | | | |
| CompACT BA | | | |
| Baseline | 17.67 (13.60) | 5.35 (4.53) | U = 24.50, p = .010 |
| Follow-up | 10.00 (5.72) | 3.88 (4.94) | |
| CompACT VA | | | |
| Baseline | 16.83 (7.70) | 11.42 (6.32) | U = 44.00, p = .100 |
| Follow-up | 7.00 (9.20) | 9.18 (6.78) | |
| CompACT OE | | | |
| Baseline | 16.83 (7.70) | 28.69 (7.99) | U = 60.50, p = .396 |
| Follow-up | 22.25 (8.85) | 23.82 (9.36) | |
| Thought intrusion | | | |
| Baseline | 7.67 (6.89) | 4.12 (4.32) | U = 48.00, p = .144 |
| Follow-up | 4.50 (4.12) | 5.18 (4.82) | |
| Cancer self-efficacy | | | |
| Baseline | 79.50 (15.76) | 93.12 (13.06) | U = 37.50, p = .050 |
| Follow-up | 93.00 (10.89) | 96.41 (8.65) | |

Note: baseline comparisons were computed using Mann Whitney-U tests given small samples.

| | | df | df _{error} | F | р | n^2_{pa} | M (SD) Control | M (SD) CBT |
|------------------------------|----------------|----|---------------------|------|------|------------|---|--|
| | | | | | | | | |
| Thought intrusion | Time | 1 | 19 | .24 | .630 | .01 | Baseline: 3.59 (4.18) Follow-up: 5.18 (3.82) | Baseline: 5.00 |
| | Condition | 1 | 19 | .03 | .867 | .00 | | (2.83) |
| | Time*Condition | 1 | 19 | .08 | .359 | .05 | | Follow-up: 4.50 (4.12) |
| Self-efficacy | Time | 1 | 19 | 7.10 | .015 | .27 | Baseline: 96.53 (8.47) Follow-up: 96.41 (8.65) | Baseline: 79.75 (20.27) |
| | Condition | 1 | 19 | 3.94 | .062 | .17 | | |
| | Time*Condition | 1 | 19 | 7.36 | .014 | .28 | | Follow-up: 93.00 (10.89) |
| Psychological flexibility | Time | 1 | 19 | 7.10 | .015 | .27 | Baseline: 43.12 (10.67) Follow-up: 36.88 (12.85) | Baseline: 55.00 (12.33) Follow- up: 39.25 (16.38) |
| | Condition | 1 | 19 | 1.72 | .205 | .08 | | |
| | Time*Condition | 1 | 19 | 1.33 | .263 | .07 | | |
| | | | | | | | | |

Table 3. Results from mixed 2*2 ANOVAs for potential mediator variables.



